Research Article



Effects of physical activity on menopausal symptoms, psychosomatic factors and well-being among working women in England: A path analysis

Women's Health Volume 20: 1–12 © The Author(s) 2024 Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/17455057241290370 journals.sagepub.com/home/whe



Nestor Asiamah^{1,2}, Olajumoke B Aladenola³, Camille Cronin³, Leeni Sepp⁴ and Kirsty O'Callaghan⁵

Abstract

Background: Research to date suggests that physical activity (PA) can buffer menopausal symptoms and support wellbeing, but there is limited evidence on the link between PA and menopausal symptoms in the United Kingdom, and no study has assessed how PA affects well-being through menopausal symptoms and three psychosomatic factors (i.e. depression, anxiety and stress).

Objectives: This study investigated whether PA affects well-being through menopausal symptoms and psychosomatic factors. This study focuses on how PA influences working women with menopause.

Design: A cross-sectional design based on the STROBE (i.e. Strengthening the Reporting of Observational Studies in Epidemiology) checklist was adopted.

Methods: The participants were 324 working women in England. An online self-reported questionnaire was utilised to gather data through Qualtrics. The data were analysed with path analysis through structural equation modelling, and sensitivity analyses were performed to avoid or reduce statistical bias.

Results: PA had a negative effect on menopausal symptoms ($\beta = -0.21$; p < 0.001) but a positive effect on well-being ($\beta = 0.19$; p < 0.001). Menopausal symptoms had a negative indirect effect on well-being, but PA had a positive indirect effect on well-being through menopausal symptoms and the three psychosomatic factors.

Conclusion: PA was positively associated with well-being but negatively associated with menopausal symptoms. Menopausal symptoms may lower well-being through anxiety, depression and stress, but PA can be associated with better well-being through depression, anxiety and stress.

Plain language summary

The influence of physical activity on well-being through menopausal symptoms, depression, anxiety, and stress

Why was the study done? Studies have assessed the potential influence of physical activity on menopausal symptoms among women, but no study in the United Kingdom has examined this relationship. Although the positive effect of physical activity on well-being is well researched and documented, no study has explored how it is mediated by menopausal symptoms and psychosomatic factors (i.e. stress, anxiety, and depression). Evaluation of this potential mediation is needed to improve stakeholders' understanding of the best ways to manage menopause at work. What

⁵National Health Service Mid and South Essex Integrated Care System, Basildon, Essex, UK

Corresponding author:

Nestor Asiamah, Division of Interdisciplinary Research and Practice, School of Health and Social Care, University of Essex, Wivenhoe, Colchester, Essex CO4 3SQ, UK. Email: n.asiamah@essex.ac.uk

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

¹Division of Interdisciplinary Research and Practice, School of Health and Social Care, University of Essex, Colchester, Essex, UK

²Department of Health Services, Africa Centre for Epidemiology, Accra North, Ghana

³School of Health and Social Care, University of Essex, Colchester, Essex, UK

⁴Colchester City Council, Colchester, Essex, UK

did the researchers do? The researchers designed an online survey and used Qualtrics to distribute it among 324 working women with menopause in England. They used the appropriate statistical analysis technique to ascertain how the effect of physical activity on well-being is mediated by menopausal symptoms and the psychosomatic factors. What did the researchers find? Women who reported higher physical activity experienced fewer menopausal symptoms. Women with higher physical activity reported better well-being through fewer menopausal symptoms and lower stress, anxiety, and depression. Physical activity may help to improve well-being among working menopausal women by lowering menopausal symptoms, stress, anxiety, and depression. What do the findings mean? Working women can maintain well-being and experience less menopausal symptoms, stress, anxiety, and depression if they participate in physical activities. Organizations can manage menopause by rolling out workplace programmes encouraging employees to regularly participate in physical activities.

Keywords

physical activity, menopause, stress, depression, anxiety, physical well-being, women

Date received: 26 January 2024; revised: 11 September 2024; accepted: 23 September 2024

Introduction

Menopause is a stage of life when women stop menstruating and is characterised by symptoms such as hot flashes, night sweats, difficulty sleeping and reduced sex drive.^{1,2} Working women are a major segment of the population experiencing menopause globally. In the United Kingdom, an increasing proportion of working women experience menopause, which is of concern to stakeholders.^{3,4} Research has found that menopause is associated with psychosomatic symptoms such as depression, anxiety and stress,⁵⁻⁹ which are key factors reducing well-being among working women. Secondly, women experiencing menopause are more likely to leave their jobs owing to the foregoing psychosomatic factors and poor well-being.¹⁰⁻¹² Stakeholders, including the British Menopause Society,⁴ have therefore called for research and interventions to improve well-being among working women experiencing menopause.

Research has shown that physical activity (PA) can buffer menopausal symptoms and possibly its adverse effect on well-being.^{13–16} PA has been defined as any bodily movement produced by skeletal muscles which requires energy expenditure above 1.5 basal metabolic rate.¹⁷ PA has been measured in terms of either hours spent in physical activities or the frequency of performing these activities over a defined period.^{18–20} In this study, the latter approach is adopted to measure PA since it better suited our data collection plan and was more convenient to the participants. Thus, PA was measured as how often the individual performed moderate (e.g. walking and gardening) and vigorous (e.g. running and bicycling) physical activities.²⁰

PA can reduce or offset the adverse effect of menopausal symptoms on psychosomatic disorders,^{21–27} which signifies a nexus between PA, menopausal symptoms, psychosomatic factors and well-being. In the context of this relationship, PA can enhance well-being in working women by buffering menopause symptoms and their influence on the psychosomatic factors. Empirically testing this nexus simultaneously can improve an understanding of the role of PA in managing

menopause at work. There have been recent calls^{4,28} for research improving stakeholders' understanding of the role of PA in managing menopause at work, and this study responds to these calls by concurrently testing, for the first time, a multivariate model incorporating PA, menopausal symptoms, the above psychosomatic factors and well-being.

Testing the above multivariate model is novel for some reasons. Firstly, the test is based on a path analysis instead of a traditional structural equation modelling (SEM). A traditional SEM, by default, would provide a single estimate of the total effect of PA on well-being through menopausal symptoms and psychosomatic factors. The total effect alone does not provide information about the indirect effects of PA on well-being through individual mediating variables (i.e. menopausal symptoms and psychosomatic factors). With a path analysis, these indirect effects are estimated and interpreted. Secondly, a path analysis enables us to identify and understand implications of the indirect effects on well-being for practice. The psychosomatic factors are among the most likely experiences of menopausal women that often affect well-being and job satisfaction.^{12,29} To enhance the evidence for informing workplace management of menopause, it is necessary to ascertain whether their relationship with menopausal symptoms and well-being can be affected by PA.

This study fills major literature gaps, which include the non-availability of research testing a model incorporating PA, menopausal symptoms, psychosomatic factors and well-being concurrently. Most of the empirical evidence linking PA to menopause comes from the United States,^{13,14} and no study has tested the link between PA, menopausal symptoms and well-being in a British sample. A study analysing this relationship in a British sample is needed because lifestyle and culture in the United Kingdom may be different from other countries studied. Differences in ethnicity and culture in the United Kingdom may influence PA and menopausal symptoms. Studies have suggested that menopausal symptoms are associated with poor wellbeing, ^{13,14,30} but the influence of PA on this relationship has

not been assessed in a British sample. In the United States, PA was found to be negatively associated with menopausal symptoms,^{13,14} but the United Kingdom may present a different result owing to its unique culture. This study builds upon this evidence and provides additional information for practitioners. Finally, no study testing models like ours has utilised a sample from the general female workforce in England. Thus, this study provides more elaborate evidence based on a sample of women from the general English female workforce.

This study builds upon our programme of research that has produced key evidence from different publications. Noteworthy is our qualitative study³¹ utilising data from five countries (i.e. England, Finland, Denmark, New Zealand, Australia and United States) where women with menopause reported a need for organisational interventions aimed at alleviating the symptoms of menopause. In another study,³² working women in the United Kingdom reported a need for similar interventions, highlighting the significance of programmes aimed at encouraging PA among menopausal women. The evidence revealed a need for stakeholders to better understand the role of PA in the management of menopause and its symptoms. The current study provides part of this understanding and the evidence needed to support menopausal women. The above contributions are made with three research questions: (1) Does PA have a direct effect on menopausal symptoms? (2) Do menopausal symptoms have a direct effect on well-being? and (3) Does PA have an indirect effect on well-being through menopausal symptoms and the psychosomatic factors?

Methods

Design

A cross-sectional design consistent with the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) checklist was adopted.³³ This design involved sensitivity analysis against confounding and measures against common methods bias (CMB). Figure 1 is a flowchart of the study design.

Participants and selection

The participants were working women experiencing menopause in England with the ability to complete the survey in English. We targeted eligible women who were members of various working groups in England. To recruit the participants, a participant information sheet with the above inclusion criteria was distributed across the groups to invite eligible individuals to participate in the study. The women participated in the study voluntarily. We calculated the minimum sample size needed for the study using relevant statistics (i.e. effect size=0.2, power=0.8 and significance level=0.05) recommended.^{34,35} The sample size reached with the G*Power software on a maximum of 14 predictors was 105. To maximise the power of our tests, we aimed to collect data from more than 105 women.

Measures

We measured menopausal symptoms with the 11-item menopause rating scale adopted in whole from previous research^{36,37} with its 5 descriptive anchors (i.e. none -1; mild -2; moderate -3; severe -4; very severe -5). This scale measures psychosomatic, physical and healthrelated symptoms of menopause. It produced a satisfactory Cronbach's $\alpha = 0.87$, which signifies its internal consistency. PA was measured with the PA behaviour domain of the health-promoting behaviour scale adopted as a whole.²⁰ This tool, which is based on a 5-point descriptive anchor (i.e. never -1; sometimes -2; about half the time -3; most of the time -4 and always), measures how often the individual had performed moderate and vigorous PA in the last 30 days. This was preferred to other measures because it best supported our online data collection method. It produced a satisfactory Cronbach's $\alpha = 0.77$ in the current study.

Well-being was measured with the unidimensional 5-item World Health Organisation well-being index.³⁸ which accompanies 5 descriptive anchors (i.e. never -1; sometimes -2; often -3; very often and always). This scale measures perceived physical and mental well-being in the general population and produced an acceptable Cronbach's $\alpha = 0.88$. Stress was measured with a 10-item scale with 4 descriptive anchors (i.e. 1 - never, 2 - sometimes; often -3and very often -4) wholly taken from the literature.³⁹ This tool measures general stress in employees and was internally consistent at Cronbach's $\alpha = 0.91$. We measured depression with the 9-item Patient Health Questionnaire with its original anchors (i.e. Not at all sure -0, several days -1, over half the days -2, nearly every day -3). This tool came from a previous study²⁸ and produced satisfactory Cronbach's $\alpha = 0.84$. Anxiety was measured with the 7-item General Anxiety Disorder scale with four descriptive anchors (i.e. not at all -1; several days -2; over half the day -3; and nearly every day -4) taken from the literature.²⁸ In the current study, it produced a Cronbach's α =0.90. Supplemental Appendix A shows scales used to measure the above variables. All negative items within the scales were reverse coded to properly align them with the rest of the items. Items of the above Likert scales were added to create a composite (total) score.

Nine variables were measured as potential covariates. One of these was marital status measured as a categorical variable with two groups (i.e. married -1, and not married -2). Smoking status (i.e. never smoked -1, ex-smoker -2 and smoker -3), employment type (i.e. part-time -1, full-time -2 and both full-time and part-time -3), whether one had multiple jobs (i.e. no -1, maybe -2, and 3 - yes), whether one had flexible job hours (i.e. no -1, and yes -2)

1. A STROBE-compliant cross-sectional design. 2. Calculated minimum sample size. 3. Selection of participant women who Design and selection met the inclusion criteria. Subjective measures (Likert scales) 1. 2. An online questionnaire hosted on Qualtrics 3. Satisfactory psychometric properties Measurement and data 4. Pilot of the survey collection 5. Ethical review and clearance 1. Generation of summary statistics Assessing multivariate normal 2. distribution of the data 3. Sensitivity analyses for confounding Data analysis 4. Structural model fitting through SEM 5. Estimation of indirect effects through path analysis

Figure I. A flowchart of the study design.

STROBE: Strengthening The Reporting of Observational Studies in Epidemiology; SEM: structural equation modelling.

and self-reported health (i.e. poor - 1, and good - 2) were also measured as categorical variables. All categorical variables were coded into dummy-type variables to support our statistical analyses. Age, job tenure and education were measured as discrete variables. Age was the individual's chronological age whereas job tenure was how long (in years) the individual had worked. Education was the participant's years of schooling. The online survey used is attached as a Supplemental Appendix B.

Steps against CMB and confounding

An online self-reported questionnaire with two main parts was used to collect data. The first section of the survey presented scales or items on the main study variables (i.e. PA, menopausal symptoms, well-being and psychosomatic factors) whereas the second part captured the demographic variables. We followed previous procedures to avoid or minimise CMB.^{35,40} Scales were randomly presented in blocks separated by preambles, which included instructions for completing each section. We then applied Herman's one-factor method by assessing the factor structure of each scale through exploratory factor analysis (EFA) with maximum likelihood estimation. The EFA produced a variance <40% on the first factor (for scales with two or more factors) or on the entire scale (for scales with a unidimensional factor structure). These results signified the absence of CMB.

Survey validation

We followed previous research^{41,42} to ensure our online survey generated useful data. We made sure the survey was not excessively long and could be completed within 15 min. Clear instructions were provided for completing the survey, and Likert scales and multiple-choice questions were used to make it easier for participants to respond. Two experts reviewed the survey to correct typographical and wording errors. Finally, we piloted the online survey with 17 participants from the sample. Responses in the pilot survey produced Cronbach's α =0.78–0.94 for all scales, signifying the appropriateness of the measures. Participants had no concerns about the survey at the pilot phase of the study.



Figure 2. The statistical (ultimate) model tested. PA, physical activity.

Data collection and ethics

Data were collected with the Qualtrics software, which was more convenient for the participants and researchers. Written informed consent was also provided online. A link of the consent form was part of the survey, and participants could only complete the survey after they had read the participant information sheet and consented to participate in the study. Coordinators of data collection distributed a link to the online survey across groups of working women experiencing menopause. Each week, the coordinators reminded members of the group about the study to encourage those who had not completed the survey to do so. Data were gathered between 15 June and 31 July 2023. A total of 351 surveys were returned by the participants, but 27 of the returned questionnaires were filled halfway and were, therefore, removed. Thus, 324 questionnaires were analysed.

Statistical analysis

Data were analysed with SPSS 28 (IBM Inc., New York, United States) and Amos; SPSS was utilised to summarise the data and perform sensitivity analyses for the ultimate confounders, whereas Amos was used to perform path analysis. Two phases of data analysis were followed. In the first phase, descriptive statistics were used to summarise the data and the hierarchical linear regression analysis was used to screen for the ultimate confounders. The goal of the sensitivity analysis, which was adopted from previous research,35 was to identify measured covariates that are more likely to affect the primary relationships assessed. Thus, it enabled us to avoid incorporating covariates that would not confound our relationships into our multivariate model. Age and self-reported health were the two ultimate covariates identified through this analysis and incorporated into the multivariate model (i.e. ultimate model). Supplemental Appendix C1 shows the steps followed in this analysis. Only four categorical variables (i.e. smoking status, employment type, having multiple jobs and flexible work hours availability) had missing data, but none of them had up to 10% missing data. Hence, we performed the sensitivity analysis with the missing data following previous research.35,43

In the second stage, we fitted two multivariate structural models, namely the baseline (non-adjusted) and ultimate (adjusted) models. The ultimate model incorporated the ultimate covariates, unlike the baseline model. Figure 2 shows

the statistical (ultimate) model tested with path analysis. There were no missing items in the data used at this stage, so our analysis was based on 2,000 bias-corrected sampling iterations through maximum likelihood estimation (at 95% confidence interval). These options enabled us to estimate indirect effect sizes along paths relevant to our research questions. The estimation was done with the 'user-defined estimands' option in Amos. Supplemental Appendix C2 shows the formulae used in the estimation. In a second sensitivity analysis, we compared the effect sizes between the baseline and ultimate models. We confirmed the multivariate normal distribution of the data, which is necessary for SEM,⁴⁴ with the Mahalanobis distance test at probability values ≥ 0.182 as recommended in the literature.⁴⁵ The statistical significance of the tests was detected at a minimum of *p* < 0.05.

Results

Table 1 shows summary statistics on the study variables. About 65% (n=209) of the participants were married, whereas 64% (n=206) were full-time workers. The average age of the participants was about 60 years (mean=59.67; SD=7.2) and the average menopausal symptoms (i.e. total score from all 11 items of the menopausal rating scale) was about 25 (mean=24.76; SD=8.21). Table 2 shows the direct effects of the adjusted and non-adjusted models. In the adjusted model, PA had a significant negative effect on menopausal symptoms (β =-0.212; critical ratio=-3.89; p < 0.001) and depression (β =-0.17; critical ratio=-5.147; p < 0.001) but a positive effect on well-being (β =0.19; critical ration=4.21; p < 0.001). This result suggests that higher PA was associated with lower menopausal symptoms and depression but higher well-being.

In the adjusted model of Table 2, menopausal symptoms were positively associated with stress (β =0.57; critical ratio=12.14; p < 0.001), anxiety (β =0.43; critical ratio=9.07; p < 0.001) and depression (β =0.46; critical ratio=10.4; p < 0.001), but negatively associated with well-being (β =-0.15; critical ratio=-2.21; p < 0.05). Thus, higher menopausal symptoms were associated with higher stress, anxiety and depression, but lower well-being. Table 3 shows the indirect total effects from the adjusted and baseline models. PA had a positive indirect effect on well-being (β =-0.13; p < 0.001) whereas menopausal symptoms (β =-0.35; p < 0.001), stress (β =-0.12; p < 0.001) and anxiety (β =-0.08; p < 0.05) had negative indirect effects on well-being.

Table 4 (i.e. adjusted model) shows the indirect effects; PA had a positive indirect effect on well-being through depression (β =0.029; p < 0.001). PA had a positive indirect effect on well-being through menopausal symptoms and depression (β =0.017; p < 0.001) and a positive indirect effect on well-being through menopausal symptoms and anxiety (β =0.012; p < 0.05). Finally, PA had a positive

Table I. Summan	ry of demographic	characteristics	(n = 324).
-----------------	-------------------	-----------------	------------

Variable	Group	n/mean	%/SD
Categorical variable			
Marital status	Married	209	64.5 I
	Not married	115	35.49
Smoking status	Never smoked	211	65.12
	Ex-smoker	101	31.17
	Smoker	9	2.78
	Missing	3	0.93
Employment type	Part-time	109	33.64
	Full-time	206	63.58
	Both	6	1.85
	Missing	3	0.93
Having multiple jobs	No	283	87.35
	Maybe	6	1.85
	Yes	32	9.88
	Missing	3	0.93
Flexible hours available	No	90	27.78
	Yes	231	71.3
	Missing	3	0.93
Self-reported health	Poor	59	18.21
	Good	261	80.56
	Missing	4	1.23
	Total	324	100
Continuous/discrete variab	le		
Depression	—	16.31	5.20
Anxiety	—	12.87	4.94
Menopausal symptoms	—	24.76	8.21
Stress	—	16.47	4.99
Physical activity	—	12.82	4.96
Well-being	—	11.92	3.58
Age	—	49.67	7.20
Job tenure (years)		15.16	46.29
Education (years)	—	14.77	7.14

n and % apply to categorical variables whereas mean and SD apply to discrete or continuous variables. *n*: frequency; SD: standard deviation.

indirect effect on well-being through menopausal symptoms and the three psychosomatic factors (i.e. stress, anxiety and depression). Figure 3 shows the fitted statistical model with standardised coefficients or effect sizes. As the footnote to Table 2 suggests, both models produced satisfactory fit statistics.

Discussion

This study aimed to test the effects of PA on well-being through menopausal symptoms and three psychosomatic factors (i.e. anxiety, depression and stress). Relevant direct and indirect effects were estimated with a path analysis.

This study found a negative effect of PA on menopausal symptoms, which suggests that working women who reported a higher frequency of PA reported lower symptoms of menopause. This negative relationship might have been explained by the cumulative effect of metabolic

DV	Path	IV	Baseline model			Adjusted model		
			Standardised (β)	Critical ratio	Þ	Standardised (β)	Critical ratio	Þ
MRS	\leftarrow	PA	-0.212	-3.892	***	-0.212	-3.892	***
Stress	\leftarrow	PA	0.004	0.087	0.931	0.004	0.087	0.931
Anxiety	\leftarrow	PA	0.071	1.824	0.068	0.071	1.824	0.068
Depression	\leftarrow	PA	-0.170	-5.147	***	-0.170	-5.147	***
Well-being	\leftarrow	PA	0.190	4.206	***	0.190	4.206	***
Stress	\leftarrow	MRS	0.569	12.139	***	0.569	12.139	***
Anxiety	\leftarrow	MRS	0.429	9.068	***	0.429	9.068	***
Anxiety	\leftarrow	Stress	0.402	8.634	***	0.402	8.634	***
Depression	\leftarrow	MRS	0.463	10.396	***	0.463	10.396	***
Depression	\leftarrow	Anxiety	0.342	7.320	***	0.342	7.320	***
Depression	\leftarrow	Stress	0.057	1.324	0.185	0.057	1.324	0.185
Well-being	\leftarrow	Anxiety	-0.176	-2.649	0.008	-0.176	-2.649	0.008
Well-being	\leftarrow	Stress	-0.117	-2.042	0.041	-0.117	-2.042	0.041
Well-being	\leftarrow	MRS	-0.149	-2.211	0.027	-0.149	-2.211	0.027
Well-being	\leftarrow	Depression	-0.234	-3.199	0.001	-0.234	-3.199	0.001
PA	\leftarrow	Age	_	_	_	-0.052	-0.957	0.339
PA	\leftarrow	poor	—	—		-0.239	-4.426	***

Table 2. Direct effects from the adjusted and baseline models (n = 324).

Fit statistics for the baseline model include chi-square = 2.33 (p=0.132); GFI=0.95, TLI=0.97 and RMSEA=0.041. Fit statistics for the adjusted model include chi-square = 2.09 (p=0.243); GFI=0.98, TLI=0.99 and RMSEA=0.023. MRS: menopausal symptoms; PA: physical activity; DV: dependent variable; IV: independent variable; GFI: goodness-of-fit index; TLI: Tucker-Lewis Index; RMSEA: root mean square error of approximation.

Table 3. Indirect effects of physical activity, menopausal symptoms, stress and anxiety on well-being (n = 324).

Dependent	Independent variable				
variable	PA	MRS	Stress	Anxiety	
Indirect β: Baseline					
MRS	_		_	_	
Stress	-0.120**		_	_	
Anxiety	-0.137**	0.229**		_	
Depression	-0.127*	0.258**	0.138**	_	
Well-being	0.126**	-0.350**	-0.116**	-0.08*	
Indirect β : Adju	isted				
MRS	—	—	_	—	
Stress	-0.120**	_		_	
Anxiety	-0.137**	0.229**	_	—	
Depression	-0.127*	0.258**	0.138**	—	
Well-being	0.126**	-0.350**	-0.116**	-0.080**	

—: Not applicable; MRS: menopausal symptoms; PA: physical activity. p < 0.05. p < 0.001.

syndrome, bone mineral density and menopausal symptom severity.⁴⁶⁻⁴⁸ Our result suggests that menopausal symptoms would be fewer as the frequency of PA increases, but this phenomenon is less likely among those experiencing extremely severe menopause. It could be attributed to PA influencing hormonal, brain and nerve function in ways that lower menopausal symptoms in women experiencing menopause. Experts suggest that menopausal symptoms are due to a distortion of hormonal function by ageing of the reproductive system.⁴⁹ If so, our result may have been influenced by the way PA impacted hormones.

The above result is consistent with several studies conducted around the world. In Turkey, for instance, PA was found in a cross-sectional study to be associated with lower menopausal symptoms.⁵⁰ Several studies^{13–15} in the United States, including randomised controlled trials,⁵¹ have confirmed that PA buffers symptoms of menopause. Our evidence is, thus, consistent with previous research. Even so, it occupies an important place in the literature since this study is the first to investigate and confirm a relationship between PA and menopausal symptoms in a British sample of women from the general working population. The above result supports studies and reports^{51,52} recognising PA as one of the ways to manage menopause at work.

This study further found a negative effect of menopausal symptoms on well-being among working women, which means that higher menopausal symptoms were associated with poorer well-being. This result is supported by previous researchers^{3,4,53} who have acknowledged that poor well-being can be a consequence of menopausal symptoms experienced by women, although the literature has reported mixed findings on the link between wellbeing and menopausal symptoms.^{3,53,54} Noteworthy is the direct positive effect of PA on menopausal symptoms, which signifies that higher PA was associated with higher

Baseline model		Adjusted model	
β	95% CI	β	95% CI
0.029**	±0.037	0.029**	±0.037
-0.009	±0.027	-0.009	±0.027
0.000	±0.022	0.000	±0.022
0.000	± 0.003	0.000	±0.003
0.000	±0.004	0.000	±0.004
0.000	±0.011	0.000	±0.011
0.017**	±0.025	0.017**	±0.025
0.012*	±0.025	0.012*	±0.025
0.010	±0.032	0.010	±0.032
0.001	±0.005	0.001	±0.005
0.003**	± 0.005	0.003**	± 0.005
0.006	±0.012	0.006	± 0.012
	$\begin{tabular}{ c c c c c } \hline Baseline model \\ \hline \beta \\ \hline 0.029^{**} \\ -0.009 \\ 0.000 \\ 0.000 \\ 0.000 \\ 0.000 \\ 0.000 \\ 0.000 \\ 0.017^{**} \\ 0.012^{*} \\ 0.012^{*} \\ 0.010 \\ 0.001 \\ 0.003^{**} \\ 0.006 \end{tabular}$	$\begin{tabular}{ c c c c } \hline Baseline model & & & & \\ \hline \hline \beta & 95\% \ Cl & \\ \hline 0.029^{**} & \pm 0.037 & \\ \hline -0.009 & \pm 0.027 & \\ \hline 0.000 & \pm 0.022 & \\ \hline 0.000 & \pm 0.003 & \\ \hline 0.000 & \pm 0.004 & \\ \hline 0.000 & \pm 0.011 & \\ \hline 0.017^{**} & \pm 0.025 & \\ \hline 0.012^{*} & \pm 0.025 & \\ \hline 0.010 & \pm 0.032 & \\ \hline 0.001 & \pm 0.005 & \\ \hline 0.003^{**} & \pm 0.012 & \\ \hline \end{tabular}$	$\begin{tabular}{ c c c c c } \hline & Baseline model & Adjusted model & Adjusted model & $$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$$

Table 4. Indirect path coefficients (effects) from user-defined estimands (n = 324).

MRS: menopausal symptoms; PA: physical activity; Dep: depression; Well: well-being; Anx: anxiety; CI: confidence interval (based on 2,000 biascorrected sampling iterations); β : regression weight.

*p < 0.05. **p < 0.001.



Figure 3. Path model with standardised regression weights. PA, physical activity.

well-being. This outcome forms the basis of the indirect positive effect of PA on well-being through menopausal symptoms (see Table 3, adjusted model). In other words, menopausal symptoms mediated the relationship between PA and well-being. Based on this result, it can be inferred that women experiencing menopausal symptoms who maintain sufficient PA may experience better well-being, compared with their counterparts with no or less PA.

All three psychosomatic factors (i.e. anxiety, stress and depression) have a negative effect on well-being. Menopausal symptoms also have a positive effect on each of these factors, which suggests that women with higher symptoms of menopause reported higher stress, anxiety and depression. This evidence is consistent with previous research²¹⁻²⁷ that have reported a negative association between these psychosomatic factors and menopausal symptoms, but this study extends the evidence to date by confirming new indirect effects of PA on well-being through menopausal symptoms and the psychosomatic factors. The first example is the indirect effect of PA on well-being through menopausal symptoms and depression (see Table 4). Another example is the positive indirect effect of PA on well-being through menopausal symptoms and anxiety (i.e. see Table 4). More compelling is the positive indirect effect of PA on well-being through menopausal symptoms and all the psychosomatic factors (see Table 4). These indirect effects of PA signify that PA can have a net positive effect on well-being, though depression and anxiety experienced by the women individually have a negative effect on well-being.

Though the foregoing indirect effects are unique to this study, they extend the scope of the evidence regarding the commonly reported role of PA in the management of menopause among working women.^{13,53} These effects suggest that PA may not only be associated with lower symptoms of menopause but may also predict well-being among women experiencing stress, anxiety and depression attributable to menopausal symptoms. Depending on the consistency of our evidence with future research, especially experimental studies establishing causation between the variables, organisations and employers may consider interventions encouraging PA among working women as programmes that would not only buffer menopausal symptoms and their influence on stress, anxiety and depression but would also preserve well-being. Interventions in the United States^{13,51,55} have proven to be effective in buffering menopausal symptoms and enhancing health-related quality of life, but none have been rolled out in the United Kingdom and most countries. We, therefore, call for more interventions and research, especially studies that can enable stakeholders to assess the effectiveness of PA in reducing menopausal symptoms and preserving well-being.

Worth mentioning is the negligible influence of measured confounders or covariates (e.g. age and education) on our multivariate model as suggested by our comparison of the baseline and ultimate models. The regression weights in both models are not different, and none of the effects was confounded. This result indicates that the direct and indirect effects of PA on well-being through menopausal symptoms and the psychosomatic factors are likely not due to the variables included in this study as covariates. Yet, our cross-sectional design could not have eliminated potential confounding,⁵⁶ so future studies incorporating more relevant covariates are needed.

Of interest are the relatively large regression weights between menopausal symptoms and the psychosomatic factors, particularly stress and anxiety. This outcome was expected since menopausal symptoms as a measure include psychosomatic indicators of stress and anxiety. The regression weights ultimately represent correlations, which imply that higher symptoms of menopause are associated with higher stress and anxiety. This evidence supports previous studies^{5,6,8,9,57} in which working women attributed their high stress and anxiety levels to menopause or reported higher stress and anxiety against higher menopausal symptoms. If women felt that their stress and anxiety were due to menopause, they may value any workplace interventions that buffer their menopausal symptoms. Further to this, this study suggests that interventions enabling employees to regularly perform PA may be valued by working women with menopause and can be effective at managing menopause in workplaces.

This study has some limitations. Our cross-sectional design could not establish causation between the variables, which means our effect sizes are ideally estimates of the association between the variables. Where possible, future researchers should employ randomised controlled trials to be able to establish causation. We did not have the chance to use a probability sampling method, and our findings may not be generalised due to our non-probability sampling method. We, therefore, call for future studies utilising probabilistic representative samples. Our study was not necessarily free of response bias since it employed only subjective or self-reported measures. Future researchers are encouraged to use objective measures where appropriate and possible. PA was measured as the frequency of physical activities performed but not as duration, miles per hour, number of steps, heaviness of weights and number of repetitions. Consequently, there may be variations in the actual effort expended in exercise. Over-estimation of whether exercise was moderate or vigorous was also possible. Our definition of PA and the use of the term 'basal metabolic rate' earlier does not mean that the activities women reported reflect exercise. Unlike some studies,⁵⁸ this study did not consider the duration of symptoms in measuring menopausal symptoms. This shortcoming of the study may have affected scores of menopausal symptoms reported.

This study was the first to test a multivariate model incorporating PA, menopausal symptoms, psychosomatic factors and well-being. Previous studies had assessed this association in parts, which narrowed the evidence. It is the first in the United Kingdom to investigate whether PA can have indirect effects on well-being through menopausal symptoms and psychosomatic factors. Its cross-sectional design is robust as it includes measures against confounding and CMB. Our sensitivity analyses for the ultimate confounding enabled us to avoid or minimise statistical bias and constitute a model for future research. Finally, the study followed the STROBE checklist,⁵⁹ which means it met relevant quality criteria that might be of interest to practitioners and future researchers. Supplemental Appendix D is the completed STROBE checklist.

Conclusion

Higher menopausal symptoms were associated with higher psychosomatic symptoms in the form of higher stress, anxiety and depression. Higher menopausal symptoms were associated with poorer well-being. PA buffered menopausal symptoms, was associated with higher well-being and had an indirect positive effect on well-being through psychosomatic factors. It is concluded that menopausal symptoms had a negative direct and indirect effects on well-being among the working women but were directly and indirectly (through the psychosomatic factors) buffered by PA. Thus, PA may enable working women to maintain wellbeing despite experiencing menopausal symptoms and their associated psychosomatic symptoms. Workplace interventions aimed at encouraging PA may help manage menopause among working women.

Declarations

Ethical approval and consent to participate

This study received ethics review and clearance from the University of Essex ethics committee (number: ETH2223-1102) after the research protocol and ethical considerations were reviewed by the committee. All participants provided written informed consent before participating in the study.

Consent for publication

The participants provided written informed consent.

Author contribution(s)

Nestor Asiamah performed statistical analysis and wrote the manuscript. Olajumoke B Aladenola coordinated data collection and was part of study conceptualisation. Camille Cronin supervised the study and was part of study conceptualisation. Leeni Sepp coordinated data collection at Colchester City Council whereas Kirsty O'Callaghan coordinated data collection at the NHS. All authors read and approved the manuscript.

Acknowledgements

None.

Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This study was seed-funded by the University of Essex.

Competing interests

The authors declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Availability of data and materials

Data used for this study will be made available upon request.

ORCID iD

Nestor Asiamah (D) https://orcid.org/0000-0002-1157-6430

Supplemental material

Supplemental material for this article is available online.

References

- Martelli M, Zingaretti L, Salvio G, et al. Influence of work on andropause and menopause: a systematic review. *Int J Environ Res Public Health* 2021; 18: 1–20.
- Ota Y, Nomura K, Hirayama J, et al. Relationship between somatic symptoms with menstruation and intention to leave work among university hospital nurses in Japan: a crosssectional study. *Int Arch Occup Environ Health* 2023; 96: 155–166.
- Lambrinoudaki I, Armeni E, Goulis D, et al. Menopause, wellbeing and health: a care pathway from the European Menopause and Andropause Society. *Maturitas* 2022; 163: 1–14.
- Hamoda H and Moger S. Developing the Women's health strategy: The British Menopause Society's recommendations to the department of health and social care's call for evidence. *Post Reprod Health* 2022; 28: 13–18.
- Matsuzaki K, Uemura H and Yasui T. Associations of menopausal symptoms with job-related stress factors in nurses in Japan. *Maturitas* 2014; 79: 77–85.
- Sood R, Kuhle CL, Kapoor E, et al. Association of mindfulness and stress with menopausal symptoms in midlife women. *Climacteric* 2019; 22: 377–382.
- Terauchi M, Odai T, Hirose A, et al. Dizziness in peri- and postmenopausal women is associated with anxiety: a crosssectional study. *Biopsychosoc Med* 2018; 12: 1–7.
- Mulhall S, Andel R and Anstey KJ. Variation in symptoms of depression and anxiety in midlife women by menopausal status. *Maturitas* 2018; 108: 7–12.
- Bryant C, Judd FK and Hickey M. Anxiety during the menopausal transition: a systematic review. *J Affect Disord* 2012; 139: 141–148.
- Olajubu AO, Olowokere AE, Amujo DO, et al. Influence of menopausal symptoms on perceived work ability among women in a Nigerian University. *Climacteric* 2017; 20: 558–563.
- Geukes M, Anema JR, van Aalst MP, et al. Improvement of menopausal symptoms and the impact on work ability: a retrospective cohort pilot study. *Maturitas* 2019; 120: 23–28.
- Griffiths A, MacLennan SJ and Hassard J. Menopause and work: an electronic survey of employees' attitudes in the UK. *Maturitas* 2013; 76: 155–159.
- McAndrew LM, Napolitano MA, Albrecht A, et al. When, why and for whom there is a relationship between physical activity and menopause symptoms. *Maturitas* 2009; 64: 119–125.
- Witkowski S, Evard R, Rickson JJ, et al. Physical activity and exercise for hot flashes: trigger or physical activity and exercise for hot flashes: trigger or treatment? https://scholarworks.smith.edu/ess_facpubs (2023, accessed 10 July 2023).

- Hulteen RM, Marlatt KL, Allerton TD, et al. Detrimental changes in health during menopause: the role of physical activity. *Int J Sports Med* 2022; 44: 389–396.
- Tong C, Meng Y, Li T, et al. High levels of physical activity are associated with a reduced likelihood of depressive symptoms in postmenopausal women. *Women Health* 2023; 63: 308–318.
- Owen N, Sugiyama T, Eakin EE, et al. Adults' sedentary behavior: determinants and interventions. *Am J Prev Med* 2011; 41: 189–196.
- Tolley APL, Ramsey KA, Rojer AGM, et al. Objectively measured physical activity is associated with frailty in community-dwelling older adults: a systematic review. *J Clin Epidemiol* 2021; 137: 218–230.
- Legh-Jones H and Moore S. Network social capital, social participation, and physical inactivity in an urban adult population. *Soc Sci Med* 2012; 74: 1362–1367.
- Wang X, Yue T and Mo PKH. The associations among cognitive social factors, eHealth literacy and health-promoting behaviors in Chinese adolescents. *Health Promot Int* 2022; 37: 1–11.
- 21. Carter T, Pascoe M, Bastounis A, et al. The effect of physical activity on anxiety in children and young people: a systematic review and meta-analysis. *J Affect Disord* 2021; 285 10–21.
- 22. Salmon P. Effects of physical exercise on anxiety, depression, and sensitivity to stress: a unifying theory. *Clin Psychol Rev* 2001; 21: 33–61.
- Lok N, Lok S and Canbaz M. The effect of physical activity on depressive symptoms and quality of life among elderly nursing home residents: randomized controlled trial. *Arch Gerontol Geriatr* 2017; 70: 92–98.
- Anderson E and Shivakumar G. Effects of exercise and physical activity on anxiety. *Front Psychiatry* 2013; 4: 27.
- Bischoff LL, Otto AK, Hold C, et al. The effect of physical activity interventions on occupational stress for health personnel: a systematic review. *Int J Nurs Stud* 2019; 97: 94–104.
- Stults-Kolehmainen MA and Sinha R. The effects of stress on physical activity and exercise. *Sports Med* 2014; 44: 81–121.
- Martins LCX and Lopes CS. Rank, job stress, psychological distress and physical activity among military personnel. *BMC Public Health* 2013; 13: 716.
- Hardy C and Hunter MS. Premenstrual symptoms and work: exploring female staff experiences and recommendations for workplaces. *Int J Environ Res Public Health* 2021; 18: 3647.
- 29. Verdonk P, Bendien E and Appelman Y. Menopause and work: a narrative literature review about menopause, work and health. *Work* 2022; 72: 483–496.
- Karaçam Z and Şeker SE. Factors associated with menopausal symptoms and their relationship with the quality of life among Turkish women. *Maturitas* 2007; 58: 75–82.
- 31. Cronin C, Bidwell G, Carey J, et al. Exploring digital interventions to facilitate coping and discomfort for nurses experiencing the menopause in the workplace: an international qualitative study. *J Adv Nurs* 2023; 79: 3760–3775.
- Cronin C, Abbott J, Asiamah N, et al. Menopause at work an organisation-based case study. *Nurs Open* 2024; 11: 1–8.
- von Elm E, Altman DG, Egger M, et al. The strengthening the reporting of observational studies in epidemiology (STROBE) statement: guidelines for reporting observational studies. *Int J Surg* 2014; 12: 1495–1499.

- Brydges CR. Effect size guidelines, sample size calculations, and statistical power in gerontology. *Innov Aging* 2019; 3: 1–8.
- Sghaier S, Asiamah N, Danquah E, et al. Information technology ability mediates the association between older adults' subjective age and social activity: a STROBEcompliant cross-sectional analysis. *Arch Gerontol Geriatr* 2022; 103: 104790.
- Heinemann K, Ruebig A, Potthoff P, et al. The Menopause Rating Scale (MRS) scale: a methodological review. *Health Qual Life Outcomes* 2004; 2: 1–8.
- 37. Suarez-García I, Alejos B, Pérez-Elías M-J, et al. How do women living with HIV experience menopause? Menopausal symptoms, anxiety and depression according to reproductive age in a multicenter cohort. *BMC Womens Health* 2021; 21: 223.
- Lara-Cabrera ML, Betancort M, Muñoz-Rubilar A, et al. Psychometric properties of the WHO-5 well-being index among nurses during the COVID-19 Pandemic: a crosssectional study in three countries. *Int J Environ Res Public Health* 2022; 19: 10106.
- 39. Ng SM. Validation of the 10-item chinese perceived stress scale in elderly service workers: one-factor versus two-factor structure. *BMC Psychol* 2013; 1: 9.
- Jakobsen M and Jensen R. Common method bias in public management studies. *Int Public Manag J* 2015; 18: 3–30.
- Asiamah N, Opuni FF, Mends-Brew E, et al. Short-term changes in behaviors resulting from COVID-19-related social isolation and their influences on mental health in Ghana. *Community Ment Health J* 2021; 57: 79–92.
- Danquah E and Asiamah N. Associations between physical work environment, workplace support for health, and presenteeism: a COVID-19 context. *Int Arch Occup Environ Health* 2022; 95: 1807–1816.
- 43. Asiamah N, Lowry R, Khan HTA, et al. Associations between social support provided and walkability among older adults: health self-consciousness as a moderator. *Arch Gerontol Geriatr* 2022; 101: 104691.
- Ernst AF and Albers CJ. Regression assumptions in clinical psychology research practice – a systematic review of common misconceptions. *PeerJ* 2017; 2017: 1–16.
- 45. Asiamah N, Adu-Gyamfi K, Frimpong FKS, et al. Development of a scale measuring nurses' physical activity counseling in a primary care facility: implications for healthcare quality. *Hosp Top* 2021; 99: 119–129.
- Cengiz H, Kaya C, Suzen Caypinar S, et al. The relationship between menopausal symptoms and metabolic syndrome in postmenopausal women. *J Obstet Gynaecol* 2019; 39: 529–533.
- 47. Alay I, Kaya C, Cengiz H, et al. The relation of body mass index, menopausal symptoms, and lipid profile with bone mineral density in postmenopausal women. *Taiwan J Obstet Gynecol* 2020; 59: 61–66.
- Kaya C, Cengiz H, Yeşil A, et al. The relation among steroid hormone levels, lipid profile and menopausal symptom severity. *J Psychosom Obstet Gynecol* 2017; 38: 284–291.
- Al-Azzawi F and Palacios S. Hormonal changes during menopause. *Maturitas* 2009; 63: 135–137.
- 50. Tan MN, Kartal M and Guldal D. The effect of physical activity and body mass index on menopausal symptoms in

Turkish women: a cross-sectional study in primary care. *BMC Womens Health* 2014; 14: 38.

- 51. Elavsky S and McAuley E. Physical activity and mental health outcomes during menopause: a randomized controlled trial. *Ann Behav Med* 2007; 33: 132–142.
- Nelson LA, Noonan CJ, Goldberg J, et al. Social engagement and physical and cognitive health among American Indian participants in the health and retirement study. *J Cross Cult Gerontol* 2013; 28: 453–463.
- Bondarev D, Sipilä S, Finni T, et al. The role of physical activity in the link between menopausal status and mental well-being. *Menopause* 2020; 27: 398–409.
- Dennerstein L, Lehert P and Guthrie J. The effects of the menopausal transition and biopsychosocial factors on wellbeing. *Arch Womens Ment Health* 2002; 5: 15–22.

- 55. Elavsky S. Physical activity, menopause, and quality of life: the role of affect and self-worth across time. *Menopause* 2009; 16: 265–271.
- Asiamah N, Mends-Brew E and Boison BKT. A spotlight on cross-sectional research: addressing the issues of confounding and adjustment. *Int J Healthc Manag* 2021; 14: 183–196.
- Hardy C, Griffiths A and Hunter MS. What do working menopausal women want? A qualitative investigation into women's perspectives on employer and line manager support. *Maturitas* 2017; 101: 37–41.
- Freeman EW, Sammel MD, Lin H, et al. Duration of menopausal hot flushes and associated risk factors. *Obstet Gynecol* 2011; 117: 1095–1104.
- 59. Cuschieri S. The STROBE guidelines. *Saudi J Anaesth* 2019; 13: S31–S34.